Author’s response to reviews

Title: Predicting patient use of general practice services in Australia: Models developed using national cross-sectional survey data.

Authors:

Christopher Harrison (Christopher.Harrison@sydney.edu.au)

Joan Henderson (joan.henderson@sydney.edu.au)

Graeme Miller (graeme.miller@sydney.edu.au)

Helena Britt (helena.britt@sydney.edu.au)

Version: 2 Date: 13 Jan 2019

Author’s response to reviews:

To the Editor.

As requested, the following items have been completed:

1. Specific information about informed consent has been clarified in the 'Declarations' section and added to the Methods section (2nd paragraph).
2. A full list of abbreviations used has been added to the 'Declarations' section.
3. Figure titles and subheadings have been included after the references.
4. All sections under the 'Declarations' heading have been completed separately.
5. A clean version of the manuscript, as requested, has been uploaded.